Author's response to reviews

Title: Tumor evolution and intratumor heterogeneity of an epithelial ovarian cancer investigated using next-generation sequencing

Authors:

Jung-Yun Lee (yodrum682@gmail.com)
Jung-Ki Yoon (dr.jkyyoon@gmail.com)
Boyun Kim (romanticet@snu.ac.kr)
Soochi Kim (skim245@snu.ac.kr)
Min A Kim (everest@snu.ac.kr)
Hyeonseob Lim (limhs0915@gmail.com)
Duhee Bang (duheebang@gmail.com)
Yong-Sang Song (yssong@snu.ac.kr)

Version: 6
Date: 21 January 2015

Author's response to reviews: see over
Jan 21, 2015

Dear Editor,

We are very grateful to you and to the reviewers for the valuable comments on our manuscript. We have taken all the comments into account and submit a second revision.

We clarified the statement following the editor’s request.

This study was approved by the Institutional Review Board (IRB) at Seoul National University Hospital (Registration number: C-1305-546-487) and performed in compliance with the Helsinki Declaration.
We obtained consent for samples to be used in research
We obtained consent to publish the case report and any accompanying images from the patient in question
We obtained consent to publish and disclose sequence data.
We deposit sequence data (Sequencing data are accessible at Sequence Read Archive, (SRA, accession number SRS823287)).

Also, we added the abovementioned sentence in the manuscript (Line 127, Line 136)

I thank for constructive suggestions and look forward to having our manuscript published in the BMC Cancer.

Sincerely,

Duhee Bang
Department of Chemistry, Yonsei University, Room 437, Science Building, 50 Yonsei-ro, Seodaemun-gu, Seoul 120-749, Korea
E-mail: duheebang@yonsei.ac.kr, Phone: 82-2-2123-2633

Yong-Sang Song
Department of Obstetrics and Gynecology, Seoul National University, College of Medicine, 101 Daehak-ro, Jongno-gu, Seoul 110-744, Korea
E-mail: yssong@snu.ac.kr, Phone: 82-2-2072-2822, Fax: 82-2-762-3599